Author's response to reviews

Title: Challenges In The Pre and Postnatal Diagnosis Of Mediastinal Cystic Hygroma

Authors:

Sarfraz A Nazir (sarfraznazir@doctors.org.uk)
Syed A Raza (arsalanraza@gmail.com)
William Sherwood (wsherwood@doctors.org.uk)
Colene Bowker (colene.bowker@orh.nhs.uk)
Kokila Lakhoo (Kokila.Lakhoo@paediatrics.ox.ac.uk)

Version: 3 Date: 5 December 2007

Author's response to reviews:

Second title page has been removed.
Competing interests section added after the Conclusions.
Authors' contributions section included.
Consent section included.
Title ends with the phrase ": a case report"
Number of figures reduced to 3.
'Index words’ section removed.

All the above as requested.